CS 66.460 Introduction to Graphical Human-Machine Interfaces Spring 1997 STUDENT INFO SHEET Monday, January 13, 1997

ffl Name:

ffl E-Mail:

ffl Home Page URL:

ffl Your major and degree program:

ffl Rate your experience level for these tools using the scale shown:

Scale: 0 = None: I have never used it
       1 = Familiar: have used once or twice
       2 = Intermediate: have used it several times
       3 = Expert: I know a lot about it

C++ 0.....1.....2.....3
C 0.....1.....2.....3
XLib Programming 0.....1.....2.....3
X Toolkit Programming 0.....1.....2.....3
Tcl/tk 0.....1.....2.....3
HTML 0.....1.....2.....3
Java 0.....1.....2.....3
CGI 0.....1.....2.....3
Perl 0.....1.....2.....3
MS Windows Programming 0.....1.....2.....3

ffl What would you like to see covered in this course? Or, what do you hope to get out of this course?